Author's response to reviews

Title: Variety of transversus thoracis muscle in relation to the internal thoracic artery: an autopsy study of 120 subjects

Authors:

Lazar Jelev (ljelev@abv.bg)
Stanislav Hristov (hristov_forensic@yahoo.com)
Wladimir Ovtscharoff (ovcharov@medfac.acad.bg)

Version: 4 Date: 25 January 2011

Author's response to reviews: see over
To
Mr Vipin Zamvar
Editor-in-Chief
(Royal Infirmary of Edinburgh
Department of Cardiothoracic Surgery
Edinburgh
United Kingdom
Email: zamvarv@hotmail.com

To
Professor David Taggart
Editor-in-Chief
(Oxford Heart Centre
John Radcliffe Hospital
Oxford
United Kingdom

Dear Editors-in-Chief,
Dear Reviewers,

Thank you for the careful consideration regarding our manuscript entitled “Variety of transversus thoracis muscle in relation to the internal thoracic artery: an autopsy study of 120 subjects” (MS: 1563013720427806) by L. Jelev, S. Hristov and W. Ovtscharoff, submitted to Journal of Cardiothoracic Surgery.

Following the helpful comments, we revised our work.

Formatting changes to be made:
1/ Major revisions (we require the author to make these changes)

Tables: table 2 has been divided into parts. Please merge the tables into one larger table or alternatively, rename and cite in the text tables 2 and 3. If keeping as separate tables, please remember to provide table titles and also to change table 3 to table 4 and cite in the text as such.

WE MERGE THE TABLES INTO ONE LARGER TABLE 2.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

WE MADE THE FINAL REVISION AND CORRECTION OF THE TEXT.

Sincerely yours,
Lazar Jelev, MD, PhD